Author's response to reviews

Title: The Relation Between Neutrophil-to-Lymphocyte Ratio and Coronary Chronic Total Occlusions

Authors:

Kenan Demir (drkenan76@yahoo.com)
Ahmet Avci (drahmetavci@yahoo.com)
Bulent B Altunkeser (bbaltunkeser@hotmail.com)
Ahmet Yilmaz (dr.ahmetyilmaz@gmail.com)
Fikret Keles (keles.fikret@gmail.com)
Ahmet Ersecgin (ersecgin@gmail.com)

Version: 2  Date: 12 July 2014

Author's response to reviews:

Dear reviewers,

We correct our mistakes. And, we've completed the missing information.

Sincerely yours,

Ass.Prof. Kenan Demir